Author's response to reviews

Title: Pregnancy Associated Breast Cancer and Increased Risk of Pregnancy-Associated Recurrence

Authors:

Freya Schnabel (writejenn@gmail.com)
Jessica Billig (Jessica.billig@nyumc.org)
Arielle Cimeno (arielle.cimeno@nyumc.org)
Jennifer Chun (jennifer.chun@nyumc.org)

Version: 2 Date: 2 February 2012

Author's response to reviews:

We have revised the manuscript according to the template that was provided:
1. We included the ethnicity of the patient in the abstract and case presentation section.
2. We split the conclusion to separate discussion and conclusion sections.
3. We included a competing interest statement.
4. There are no acknowledgements.
5. We included the authors' contribution section.